Author's response to reviews

Title: Familial tumoral calcinosis in two Chinese patients: a case series

Authors:

Che Zhang (prof.zh@163.com)
Jiaowei Gu (giw888giw@163.com)
Xiaoli Chen (cattych163@163.com)
Kui Xiong (xiongkui@csc.org.cn)

Version: 5 Date: 20 March 2011

Author's response to reviews: see over
Cover Letter

MS: 2094385716459480
Familial tumoral calcinosis in two Chinese patients: a case series
Che Zhang, Jiaowei Gu, Xiaoli Chen and Kui Xiong

Comments from Associate Editor:
1. “They have to revise the Discussion and make it more crisp.”
   Response: The description about Secondary Tumoral Calcinosis and Treatment in discussion section were simplified. 2 reference were deleted.

2. “In the Introduction and in the Discussion, they should update with details of the more recent work on the subject.”
   Response: The subjects lived in countryside and were lost to follow up. We did all the efforts to get in touch with them via telephone and address, but failed to get further information. No further study on it would be conducted recently due to difficult subject collection.